Author’s response to reviews

Title: Acute abdominal pain in a patient on automated peritoneal dialysis - acute appendicitis: a case report

Authors:

Robert Ekart (robert.ekart2@guest.arnes.si)
Matjaž Horvat (maco.horvat@gmail.com)
Miran Koželj (miran.kozelj@ukc-mb.si)
Breda Pečovnik Balon (breda.balon@ukc-mb.si)
Sebastjan Bevc (sebastjanb@yahoo.com)
Radovan Hojs (radovan.hojs@ukc-mb.si)

Version: 2 Date: 29 February 2012

Author’s response to reviews:

I have checked the manuscript, include the study design in the title and also, I can confirm, that I have obtained written informed consent from the patient’s next of kin for publication of the case.